Author’s response to reviews

Title: Low dose of mycophenolate mofetil is enough in desensitized kidney transplantation using rituximab

Authors:

Chung Hee Baek (bch393@naver.com)
Hyosang Kim (mateus111@daum.net)
Hoon Yu (yuhoon721@gmail.com)
Eunhye Shin (seh82@naver.com)
Hyungjin Cho (yourhyungjin@hanmail.net)
Won Seok Yang (wsyang@amc.seoul.kr)
Duck Jong Han (djhan@amc.seoul.kr)
Su-Kil Park (skpark@amc.seoul.kr)

Version: 1 Date: 13 Nov 2015

Author’s response to reviews:

Response to the comments of Reviewer #1

In this manuscript, Baek and Colleagues evaluated the safety of low dose of mycophenolate mofetil (MMF) in HLA-sensitized or AB0 incompatible living donor kidney transplantation. They found that a lower dose of MMF was effective in reducing infectious complications but acute rejection occurred at the same rate in all the three groups. Graft survival was not different between low and standard dose of MMF, as well as patients’ survival. This paper is interesting and could be published after an extensive revision of the English language.

We performed extensive revision of the English language with native speakers. We indicated the changes in the manuscript using track changes.

Response to the comments of Reviewer #3

I appreciated your work in a field that always needs to be implemented to the best of our knowledge and possibilities. I think that results and limitations of the work are well explained; I just wasnt you to notice a text error in line 191 (antibody-mediCated instead of antibody-mediated)
We corrected the text error in line 191 (antibody-mediCated  antibody-mediated) and indicated the change using red color (the line 191 was changed to the line 206 after revision). Thank you for your comment.